Author's response to reviews

Title: A walking programme and a supervised exercise class versus usual physiotherapy for chronic low back pain: a single-blinded randomised controlled trial. (The Supervised Walking In comparison to Fitness Training for Back Pain (SWIFT) Trial).

Authors:

Deirdre A Hurley (deirdre.hurleyosing@ucd.ie)
Grainne O'Donoghue (grainne.odonoghue@ucd.ie)
Mark A Tully (m.tully@qub.ac.uk)
Jennifer Klaber Moffett (J.K.Moffett@hull.ac.uk)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Leslie Daly (leslie.daly@ucd.ie)
Colin AG Boreham (colin.boreham@ucd.ie)
Suzanne M McDonough (s.mcdonough@ulster.ac.uk)

Version: 2 Date: 28 May 2009

Author's response to reviews:

The revised version of the manuscript contains the abstract and authors list within the main manuscript. They were originally submitted as separate files.